University of Montana

ScholarWorks at University of Montana

Biological Sciences Faculty Publications

Biological Sciences

2-2021

Evolutionary dynamics and structural consequences of de novo beneficial mutations and mutant lineages arising in a constant environment

Margie Kinnersley University of Montana

Katja Schwartz Stanford University School of Medicine

Dong Dong Yang Georgia Institute of Technology

Gavin Sherlock Stanford University School of Medicine

Frank Rosenzweig University of Montana

Follow this and additional works at: https://scholarworks.umt.edu/biosci_pubs

Part of the Biology Commons Let us know how access to this document benefits you.

Recommended Citation

Kinnersley, M., Schwartz, K., Yang, DD. et al. Evolutionary dynamics and structural consequences of de novo beneficial mutations and mutant lineages arising in a constant environment. BMC Biol 19, 20 (2021). https://doi.org/10.1186/s12915-021-00954-0

This Article is brought to you for free and open access by the Biological Sciences at ScholarWorks at University of Montana. It has been accepted for inclusion in Biological Sciences Faculty Publications by an authorized administrator of ScholarWorks at University of Montana. For more information, please contact scholarworks@mso.umt.edu.

RESEARCH ARTICLE

Open Access

Evolutionary dynamics and structural consequences of de novo beneficial mutations and mutant lineages arising in a constant environment



Margie Kinnersley^{1†}, Katja Schwartz^{2†}, Dong-Dong Yang³, Gavin Sherlock^{2*} and Frank Rosenzweig^{1,3*}

Abstract

Background: Microbial evolution experiments can be used to study the tempo and dynamics of evolutionary change in asexual populations, founded from single clones and growing into large populations with multiple clonal lineages. High-throughput sequencing can be used to catalog de novo mutations as potential targets of selection, determine in which lineages they arise, and track the fates of those lineages. Here, we describe a long-term experimental evolution study to identify targets of selection and to determine when, where, and how often those targets are hit.

Results: We experimentally evolved replicate *Escherichia coli* populations that originated from a mutator/nonsense suppressor ancestor under glucose limitation for between 300 and 500 generations. Whole-genome, whole-population sequencing enabled us to catalog 3346 de novo mutations that reached > 1% frequency. We sequenced the genomes of 96 clones from each population when allelic diversity was greatest in order to establish whether mutations were in the same or different lineages and to depict lineage dynamics. Operon-specific mutations that enhance glucose uptake were the first to rise to high frequency, followed by global regulatory mutations. Mutations related to energy conservation, membrane biogenesis, and mitigating the impact of nonsense mutations, both ancestral and derived, arose later. New alleles were confined to relatively few loci, with many instances of identical mutations arising independently in multiple lineages, among and within replicate populations. However, most never exceeded 10% in frequency and were at a lower frequency at the end of the experiment than at their maxima, indicating clonal interference. Many alleles mapped to key structures within the proteins that they mutated, providing insight into their functional consequences.

Conclusions: Overall, we find that when mutational input is increased by an ancestral defect in DNA repair, the spectrum of high-frequency beneficial mutations in a simple, constant resource-limited environment is narrow, resulting in extreme parallelism where many adaptive mutations arise but few ever go to fixation.

Keywords: E. coli, Adaptation, Experimental evolution, Clonal interference, Parallelism, Resource limitation

frank.rosenzweig@biology.gatech.edu

¹Margie Kinnersley and Katja Schwartz contributed equally to this work. ²Department of Genetics, Stanford University School of Medicine, Stanford,

CA 94305-5120, USA

¹Division of Biological Sciences, The University of Montana, Missoula, MT 59812 USA

Full list of author information is available at the end of the article



[©] The Author(s). 2021 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, with http://creativecommons.org/licenses/by/4.0/. The Creative Commons.Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

^{*} Correspondence: gsherloc@stanford.edu;

Author summary

Microbial evolution experiments open a window on the tempo and dynamics of evolutionary change in asexual populations. High-throughput sequencing can be used to catalog de novo mutations, determine in which lineages they arise, and assess allelic interactions by tracking the fate of those lineages. This approach, adaptive genetics, makes it possible to discover whether clonal interactions are antagonistic or synergistic, and complements genetic screens of induced deleterious/loss-of-function mutants. Using glucose-limited chemostats, we carried out 300-500 generation evolution experiments founded by an Escherichia coli mutator/nonsense suppressor strain. Whole-genome, whole-population sequencing enabled us to catalog 3346 de novo mutations that reached >1% frequency. Mutations enhancing glucose uptake rose to high frequency first, followed by global regulatory changes that modulate growth rate and assimilation of the limiting resource; later selected mutations favored energy conservation and/or mitigated pleiotropic effects of earlier regulatory changes. Several loci were highly polymorphic, with identical mutations arising independently in different lineages, both between and within replicate populations. When mutational input is increased by an ancestral defect in DNA repair but mitigated by a nonsense suppressor, the number of beneficial mutants attributable to loss-of-function involves fewer nonsense mutations relative to missense mutations. Under nutrient limitation, selection is called upon to explore sequence space for changes in protein structure that favor, for example, de-repression of genes and pathways needed to acquire the limiting nutrient. The net result of this process is extreme parallelism, where many adaptive mutations arise within a relatively small set of genes, but few of those mutations ever become fixed. The distribution of such alleles in sequence space is useful for adaptive genetics-driven studies of protein structure-function relationships.

Background

Experimental microbial evolution has enlarged our understanding of the tempo and mode of adaptive change in asexual populations, as well as how selection, drift, and historical contingency influence their evolutionary trajectories. Using high-throughput sequencing, we can now identify substantial numbers of de novo beneficial mutations in laboratory populations, determine in which lineages they arise and the fate of those lineages, and evaluate how different alleles interact [1–3]. This approach, adaptive genetics, based on analyzing cohorts of spontaneous beneficial mutations to determine how their frequencies fluctuate over time, complements traditional genetic screening of induced deleterious/loss-offunction mutants (e.g., [4, 5]). Adaptive genetics also opens up new ways to discover constraints on protein structure and function and to discern the architecture and malleability of networks that regulate nutrient sensing and uptake and that coordinate cell division.

Microbial populations were once thought to evolve by the periodic selection of adaptive clones, each fitter than its antecedent, replacing one another over successive generations [6-9]. This model is consistent with Muller and Haldane's view of how beneficial mutations spread in large asexual populations [10-12] governed by competitive exclusion [13]; indeed, periodic selection has been observed in nosocomial outbreaks [14] and epidemics [15], as well as in breast cancer [16] and tumorspecific T cells [17]. But clonal populations can also accumulate and retain genetic variation, much of which is beneficial [18-22]. In fact, whole-genome, wholepopulation sequencing has shown that even under simple laboratory conditions the amount of adaptive genetic variation arising in microbial populations can be enormous, owing to their large size and to the continuous input of neutral and adaptive mutations [20, 23].

When novel beneficial mutations arise in independent lineages and these lineages have similar fitness, a 'Battle Royale' ensues, producing clonal interference [18, 20, 24-26]. Clonal interference can also occur within a broader framework of stable subpopulation structure [27], especially if lineages come under balancing selection [28-31] or specialize to exploit niches created by the culture conditions [29, 32, 33] or by the organisms themselves [34–36]. Theory indicates that in a resourcelimited environment the likelihood that subpopulations co-exist depends on the input of the primary resource, the output of secondary resources, and the relative fitness of clones that can profit from secondary resources [37]. The ancestral genotype may also be decisive. Glucose-limited evolution experiments carried out by Ferenci et al. using one ancestral E. coli K12 derivative never produced stable subpopulations [38], whereas those carried out by Adams et al. using another often did [36, 39]. Adams' strain was later shown to carry nonsense mutations in mismatch repair enzyme MutY, housekeeping and stationary-phase transcription factors RpoD and RpoS, as well as a tRNA nonsense suppressor. While an ancestral defect in DNA repair would increase the descendant population's mutational load [34, 40], a nonsense suppressor in that ancestor would likely mitigate the effect of any mutation that caused a premature STOP codon. This genotype could be expected not only to increase the overall number of mutations, but also the number of mutations whose beneficial effects can be traced back to structural changes in regulatory genes, especially those that encode repressor proteins, giving insight into the function of those regions whose structure has been altered.

To understand the impact that a mutator/suppressor founder has on the spectrum and fate of new beneficial mutations and on the dynamics of population structure, we repeated the classic Adams et al. experiments using the same ancestral strain and culture conditions [36]. Over the course of up to 500 generations, we monitored, at 50-generation intervals, the incidence of mutations that reached at least 1% frequency, identifying both beneficial and hitchhiking mutations. To determine which mutations co-occurred within a given lineage, we sequenced 96 clones from each population at the time point where we observed the greatest allelic diversity. We uncovered no evidence for stable subpopulation structure, but instead observed pervasive clonal interference, with only 17 out of 3346 mutations (of which a few hundred are likely beneficial) going to near fixation across replicate experiments. The temporal order in which certain mutations rose to high frequency was largely predictable, reflecting a high degree of parallelism among replicates. In general, mutations that enhanced glucose assimilation arose early, followed by mutations in global regulators and mutations that either increased efficiency of limiting resource utilization or mitigated the deleterious effects of certain earlier mutations. Our results show that when replicate populations of mutatorsuppressor E. coli evolve under carbon limitation the number of allelic variants that exceed 1% frequency may be large, but the number of genes targeted is relatively few. We further show that in many cases the distribution of high-value mutations is clustered in regions essential for gene products to exert their regulatory function.

Results

Experimental design

Evolution experiments were carried out in triplicate under continuous nutrient limitation using Davis Minimal Medium [36], with glucose (0.0125% w/v) as the sole source of carbon for energy and growth. In addition to archiving samples as - 80 °C glycerol stocks, experimental populations were also monitored every 10-20 generations for culture purity by microscopy and by plating cultures onto a lawn of multiple E. coli-specific bacteriophage, as previously described by [41]. Chemostats (300 mL working volume) were run under aerobic conditions for 300-500 generations at constant temperature (30 °C) and at constant dilution rate (D = $0.2 h^{-1}$). Under these conditions, steady-state population density is $\sim 10^8$ cells mL⁻¹ and residual glucose concentration is at or below the limit of detection (Additional file 1: Fig. S1). The E. coli strain used to initiate these experiments, JA122, is distinguished from E. coli K12 by alleles likely to influence the spectrum of mutations arising during adaptive evolution (Additional file 2: Table S1 [34];). Among these is a nonsense mutation in MutY (Leu299*) that results in a mutation rate nearly 30-fold greater than K12 [34], nonsense mutations in genes that encode stationary-phase sigma factor RpoS (Gln33*) [42], and 'housekeeping' sigma factor RpoD (Glu26*), as well as a suppressor mutation in the *glnX* tRNA known to suppress amber, ochre, and opal mutations (Additional file 2: Table S1) [43].

To identify the mutations that arose during our experiments, we performed whole-genome, whole-population sequencing every 50 generations on each of the three chemostat populations. We generated approximately 50 million 2×100bp paired end reads per sample, yielding coverage of up to $\sim 1000x$ for each time point (library insert sizes were selected to be short enough such that forward and reverse reads overlapped, which, while reducing coverage, increases quality; see the "Methods" section). We used these data to identify mutations that rose to an allele frequency of ~ 1% or greater. Given an effective population size of $> 10^{10}$ and 300–500 generations of selection, it is highly improbable that any allele could reach such a frequency by drift alone [29]. We therefore assume that every mutation we identified had either come under positive selection or was hitchhiking along with one that had.

Population sequencing reveals consistent mutation patterns across independent evolution experiments

Across all samples, 3346 SNPs were detected in 2083 unique genes or intergenic regions (Additional file 3: Table S6). The overwhelming majority (97.5%) of these SNPs were $GC \rightarrow TA$ transversions, as expected given the ancestral strain's defect in mismatch repair protein MutY, which encodes adenine glycosylase [44]. Consistent with the protein coding density of E. coli (87.8%) [45], 85% (2854) of SNPs occurred in coding regions. On average, 69.2% of these created a missense mutation, 23.4% resulted in a synonymous mutation, and 7.4% caused a nonsense mutation (Additional file 1: Fig. S2). Relative to proportions observed in mutation accumulation experiments carried out using wild-type E. coli [46], we observed more nonsynonymous and nonsense mutations. Given that MA experiments deliberately avoid selection pressure, through single-cell bottlenecks, while evolution experiments typically purge highly deleterious mutations, the greater fraction of nonsense mutations (7.4% vs. 3% in [46]) observed in our experiment is all the more striking. However, we note that a more appropriate comparison would be to compare a suppressed vs. a non-suppressed mutator strain to determine if an excess of nonsense mutations occurs as a result of suppression; this would require the existing, suppressed mutations in the background be reverted.

Small deletions were rarely detected (one singlenucleotide deletion in each of vessel 1 and vessel 2, and none detected in vessel 3), but we observed a single large ~ 150 kb duplication in vessel 2. The overall number of mutations in each population increased linearly over time and at approximately the same rate across replicates (Additional file 1: Fig. S2), as would be expected with a mutator phenotype.

Comparison of population-level mutations reveals clonal interference and widespread parallelism

Despite the large number of SNPs detected across replicate populations, only 17 novel alleles ever approached fixation by exceeding 98% frequency. Moreover, the maximum frequency of most alleles never exceeded 10% (Additional file 1: Fig. S3A), and the vast majority of alleles were present at a lower frequency in the final time point than they were at their maximum (Additional file 1: Fig. S3B), as has been previously observed in population sequencing data [25]. Together, the foregoing observations suggest that in each evolution experiment population dynamics was largely driven by clonal interference [47]. A small number of loci were recurrently mutated above what would be expected by chance, indicating that variants at these loci were likely beneficial (Table 1, Additional file 2: Table S2). For example, a total of 212 mutations arose in the 10 most significantly mutated genes identified in the population sequencing data, with each gene receiving at least five mutations (Table 1). Moreover, 30 and 14 distinct allelic variants were discovered in just two: the genes encoding the DNA-binding repressor GalS and the RNA-binding protein Hfq, respectively (Additional file 2: Table S3). High-resolution population sequencing also revealed that 13 SNPs not present at the start of the experiment reached at least 1% frequency in all three vessels at various time points, while 52 SNPs recurred in two out of three chemostats (Additional file 2: Table S4). Thus, our data also provide compelling evidence for substantial parallel evolution at the genic level—indeed, with only two exceptions, genes containing beneficial mutations (as determined from the population sequencing) were mutated in either two or three of the chemostats (Additional file 1: Fig. S3C).

Clonal sequencing further clarifies lineage relationships and parallelism

To establish linkage relationships between novel alleles, we sequenced 96 individual clones from each vessel. In each case, the 96 clones were isolated at random from the time point at which we detected the greatest number of mutant alleles at $\geq 5\%$ frequency. To assess whether the frequency estimates from population sequencing were reasonable, and whether the isolated clones constituted a reasonable subsample, we compared frequencies of mutations identified in both datasets at the

corresponding time point and found that they correlated well (Additional file 1: Fig. S4).

For each set of 96 clones, we constructed a phylogeny to represent their putative evolutionary relationships (Fig. 1). Inspection of the mutations and phylogenetic trees from each vessel (i.e., each independent evolution) revealed several instances in which exactly the same mutation arose not only in different vessels, but often more than once in the same vessel on distinct branches of a given tree. In the most extreme case, 6 of the 11 hfq alleles detected via clone sequencing were identified in clones from different vessels, indicating independent parallel origins (Fig. 1, Additional file 4: Table S7). Furthermore, 7 of the 11 appear to have arisen more than once within the same vessel.

Clonal dynamics are shaped by relationships among de novo alleles, hard and soft selective sweeps, and the absence of periodic selection

Combining population allele frequency data with linkage information inferred from clone sequencing makes it possible to depict lineage dynamics using Muller diagrams (Fig. 2, Additional file 5: Fig. S9, Additional file 6: Fig. S10, Additional file 7: File Fig. S11; NOTE: Additional files 5, 6, and 7 each contain a PDF with scrollable panels that depict evolutionary dynamics for > 50individual genes). In general, we observed early, hard sweeps of highly beneficial mutations related to limiting nutrient influx, followed by soft sweeps [48-50] and multiple-origin soft sweeps that may fine-tune glucose uptake or utilization later in the experiment when diversity was higher [51-53]. Hard sweeps consistently involved mutations in regulators (galS in chemostat 1, transcriptional terminator rho in chemostats 1 and 3) or regulatory regions (upstream of *dnaG* in chemostat 1, upstream of mglB in chemostats 1, 2, and 3), while soft sweeps were comprised of both regulatory and operonspecific mutations (e.g., hfq and opgH in chemostats 1, 2, and 3, upstream of adhE in chemostat 1, pgi in chemostat 3) (Figs. 2 and 3, Additional file 5: Fig. S9, Additional file 6: Fig. S10, Additional file 7: Fig. S11 [49, 54]). Here, we note that multiple-origin soft sweeps may be especially prevalent in our experiments due to the ancestral mutator allele at *mutY*, as the likelihood of concurrent identical mutations in the same gene should increase with mutation rate.

Early sweeps occur in genes that regulate influx of the limiting nutrient glucose

For specific growth rates between $\sim \mu = 0.1 \text{ h}^{-1}$ and $\mu = 0.9 \text{ h}^{-1}$, glucose is most efficiently transported using a combination of the maltoporin LamB and the galactose transporter MglBAC, and glucose limitation tends to select for mutations that increase expression of these

Table 1 Characteristics of frequently mutated genes

Rank	Gene	GO biological process	Observed mutations	Unique alleles	Expected mutations	Uncorrected <i>p</i> value	FDR
Popula	ation sequencing						
1	galS***	Regulation of transcription	38	30	0.78	6.55E-50	4.42E 45
2	hfq*******	Regulation of Translation	24	14	0.23	6.91E-40	2.33E 35
3	pgi*******	Glycolytic process	35	24	1.23	4.54E-38	1.02E 33
4	opgH**	Response to osmotic stress	31	29	1.90	8.74E-27	1.48E 22
5	malT********	Regulation of transcription	30	19	2.02	8.10E-25	1.09E 20
6	malK******	Carbohydrate transport	22	14	0.83	7.47E-24	8.40E 20
7	upstream mglB**	Transcription regulatory region	7	4	0.21	2.91E-09	2.81E 05
8	rho**	Transcription termination	11	9	0.94	5.49E-09	4.64E 05
9	upstream dnaG	Transcription regulatory region	5	5	0.08	3.06E-08	2.30E 04
10	fimH***	Cell adhesion/biofilm formation	9	5	0.68	4.38E-08	2.96E 04
Clonal	sequencing						
1	hfq******	Regulation of translation	26		0.1020	3.79E-53	3.68E 48
2	pgi****	Glycolytic process	17		0.5448	5.52E-20	2.68E 15
3	opgH***	Response to osmotic stress	17		0.8400	6.57E-17	2.13E 12
4	upstream mglB**	Transcription regulatory region	8		0.0925	1.22E-13	2.96E 09
5	fimH****	Cell adhesion/biofilm formation	10		0.2982	1.17E-12	2.26E 08
6	ompR***	Regulation of transcription	8		0.2377	2.05E-10	3.31E 06
7	upstream adhE*	Transcription regulatory region	6		0.1575	1.85E-08	0.000257
8	malT***	Regulation of transcription	10		0.8935	3.98E-08	0.000482
9	proQ*	Posttranscriptional regulation	6		0.2308	1.72E-07	0.001858
10	pfkA**	Glucose catabolic process	6		0.3180	1.09E-06	0.010612

Each asterisk indicates an allele that arose more than once independently, either within or between vessels

proteins [52, 55–63]. Accordingly, 7 of the top 10 frequently mutated genes/gene regions we observed (*galS*, upstream *mglB*, *malT*, *malK*, *hfq*, *rho*, and upstream *dnaG*) play a role in transcriptional regulation of LamB or MglBAC, either directly or through their interactions with global regulators (Table 1, Fig. 4).

Functional attributes and evolutionary dynamics of mutations in operon-specific regulators galS and mglB

When *E. coli* is cultured under glucose limitation, the chief route by which limiting substrate enters the

cytoplasm is via the D-galactose/methyl- β -D-galactoside transporter MglBAC (Fig. 4 [62]). GalS is a negative regulator of *mglBAC* transcription, and in the absence of D-galactose, GalS binds the *mgl* operator to prevent open complex formation [64]. Loss-of-function mutations in *galS* or mutations upstream of *mglB* in the GalS repressor binding site (bp 2,238,647 C \rightarrow A) and/or the CRP activator site (bp 2,238,630 C \rightarrow A) have been previously observed in the early stages of a daptation to limiting glucose [40, 58, 65, 66]. In our experiments, we observed 33 mutations in *galS*, far exceeding what would be expected by chance.



When all 33 galS mutations are mapped onto the primary protein sequence, many occur in both the helixturn-helix portion of the DNA-binding domain (aa 4-23, gray shading in Fig. 5a) and in two distinct regions of the C-terminus (aa 211-252, stipple in Fig. 5a). Although no crystal structure is available for GalS, a homology model based on the PurR repressor shows that the spatial distribution of these three groups of mutations is consistent with their placement in the DNA-binding region, dimer stabilization region, and intramolecular signaling region, respectively (Fig. 5b) [67]. Moreover, the highest frequency mutation from chemostat 1 (Arg146Leu, also detected in chemostats 2 and 3) occurred in a conserved residue near or in the presumptive galactose-binding site (Fig. 5b) [68]. All are expected to result in loss of GalS function and consequently to enhance transcription of mglBAC.

Despite their early increase in frequency, few galS mutations persisted beyond generation 50 or attained a frequency greater than 5%. Instead, the majority of galS mutants was rapidly displaced by clones carrying highly beneficial mutations in the mgl operator sequence upstream of mglB (Figs. 2, 3, and 5a). The most successful mutation upstream of *mglB* (bp 2,238,647 C \rightarrow A) occurred in every vessel and increased in frequency to > 90% (Table S2, Additional file 5: Fig. S9, Additional file 6: Fig. S10, Additional file 7: Fig. S11). Exactly the same mutation has been observed in previous E. coli chemostat evolution experiments, demonstrating the enormous benefit this allele confers under glucose limitation, regardless of nuances afforded by chemostat setup, dilution rate, or strain [40, 52, 58]. Over the remainder of the experiment, only three other mutations upstream of mglB reached the threshold for detection: two were



chemostats 1, 2, and 3, respectively

within 2 base pairs of the first mutation and did not rise to high frequency, while the third (chemostat 1, 2,238,630 C \rightarrow A), located in the CRP activator binding site, co-occurred with 2,238,647 C \rightarrow A and increased to ca. 80% frequency by generation 500 (Figs. 2 and 3a, Additional file 4: Table S7). These findings suggest additional mutations that affect GalS repressor binding are not of great benefit after the preferred allele has swept the population, whereas mutations that modulate the activity of other regulators (i.e., CRP) can act synergistically.

Functional attributes and evolutionary dynamics of mutations in genes that directly regulate LamB expression: malT and malK

Increased expression of the gene encoding outer membrane glycoporin LamB is another hallmark feature of *E. coli* adapted to glucose-limited chemostat growth [40, 52, 57, 59, 60, 69]. Previous experiments have shown that under glucose limitation, LamB overexpression can result from any one of the following: constitutive activation of transcriptional regulator MalT, disruption of the MalT inhibitor MalK, mutation of the RNA chaperone



Hfq, alteration of sigma factor dynamics (σ^{S}/σ^{D} ratio), or mutation of the *malT* repressor Mlc (Fig. 4) [40, 52, 57– 60, 69, 70]. Across replicate experiments, we observed 19 unique *malT* alleles and 14 unique *malK* alleles (Fig. 6, Additional file 1: Fig. S5, Additional file 2: Table S3). Over half of the *malT* mutations (10 out of 19) are known either to cause MalT to become constitutively active, or to occur in amino acids involved in MalT/ MalK interaction (Fig. 6a) [52, 71, 72]. Likewise, the majority of MalK substitutions (10 out of 14 different alleles, Fig. 6a) occurred within or just outside of the MalK/MalT interaction domain and are predicted to weaken MalT inhibition [73, 74].

The diversity of *malT* mutations reflects the diversity of signals integrated by MalT. All were missense substitutions, and all fell roughly into four clusters that correspond to a nucleotide-binding domain (aa cluster 4-60), a linker region/winged-helix domain (aa clusters 236-358 and 311-319), and a small patch of the maltotriose sensor domain (aa cluster 634-637). Mutations in the nucleotide-binding domain and winged-helix domain almost precisely delineate regions of the primary sequence previously identified by Schlegel et al. [75] as associated with a *mal* constitutive phenotype (Fig. 6a, red shading) [52, 71, 72]. The fourth cluster of mutations in the sensor domain (Arg634Ser/Arg634Leu/Asn637Lys) is on the surface of MalT in a 7 amino acid stretch of DT3 that serves as a point of contact between MalT and MalK (Fig. 6b) [71]. Mutations in this region eliminate MalK inhibition, thereby increasing transcription of mal genes. We recovered no mutations in the DNA-binding domain of MalT (effector domain), consistent with the expectation that our adaptive variants retain the ability to activate MalT-responsive promoters [71, 76, 77].

Ten of 14 substitutions in MalK occurred in the Cterminal 2/5 of the protein and were primarily located in or underneath predicted MalT binding sites (Fig. 6c) [73, 74]. Mutations in this region affect MalT inhibition and promote the expression of *mal* genes, but do not generally affect maltose transport, reflecting the dual role and two-domain nature of MalK [73, 78]. Comparatively few mutations (4 out of 14 alleles) occurred in the N-terminal nucleotide-binding domain. However, amino acid 51 was mutated three times over during the course of the experiment and is situated adjacent to proline 72, modification of which has also been shown to decrease MalK regulatory activity (Fig. 6d) [73, 74].

The dynamics of *malT/malK* allele frequency differs among experimental populations. In chemostat 1, MalK Ala296Asp rapidly sweeps to fixation, whereas in chemostat 3, early mutations in MalK (yellow) are displaced by later mutations in MalT (purple) (Fig. 3a, c). In chemostat 2, clones with either malK or malT mutations co-exist through all 500 generations (Additional file 1: Fig. S5). The reason for this contrast in dynamics cannot be attributed to emergence of a single "most fit" allele, as the majority types from chemostats 1 and 3 arose independently in chemostat 2, but did not sweep. Despite the fact that MalT and MalK are high-value targets of selection during adaptation to glucose limitation, other advantageous mutations (upstream mglB, rho, and hfq, discussed below) may have ultimately carried "winning" mal alleles in chemostats 1 and 3 to higher frepurging allelic diversity at this locus. quency,



Interestingly, although we observed 30 *malT* and 22 *malK* mutations in the population sequencing data (Table 1), with mutations being observed in 117 and 116 of the sequenced clones respectively, in only 5 out of the 288 sequenced clones do mutant alleles of these two genes co-occur, suggesting that there may be little or no additional advantage or even some disadvantage (due to reciprocal sign epistasis [79]) to having both. A lack of co-occurrence *of malT* and *malK* mutations has also been observed in previous evolution experiments [36], in which the primary resource (glucose) specialist carries a mutation in MalK and the secondary resource specialists share a mutation in MalT [40].

Functional attributes and evolutionary dynamics of mutations in global regulators that enhance glucose assimilation: Hfq, rho, and the t1 terminator

Hfq is a global regulatory protein that facilitates translation and/or RNA degradation by mediating ncRNA- mRNA interactions. It participates in a wide range of cellular processes including nutrient uptake, motility, and metabolism and is also a general regulator of stress response via interactions with mRNAs that encode sigma factors σ^{S} , σ^{E} , and σ^{H} [80, 81]. *hfq* mutations identified in other glucose-limited evolutions were found to be pleiotropic, increasing translation of LamB glycoporin, reducing levels of stationary-phase transcription factor RpoS, inhibiting cellular aggregation, and enhancing glucose transport via PtsG [70, 82].

Hfq is one of the most frequently mutated genes in our experiments: 24 *hfq*-independent mutations were recovered by population sequencing, comprising 14 unique *hfq* alleles; when the experiments were terminated, > 50% of cells in each population carried an *hfq* mutation (Additional file 2: Table S3; Table 1). Two of these alleles arose independently in all three vessels (same nucleotide position, same SNP), and six additional alleles were observed in two of three vessels (Additional file 2:



involved in dimer stabilization (green). The PurR ligand guanine is shown in gray cartoon style (PDBID 1WET) [67]. Middle: SWISSMODEL representation of the GalS repressor based on the structure of PurR (PDBID 1JFS, 32.53% sequence identity). Mutations grouped in the N-terminal DNA-binding domain are shown as orange spheres, while the two groups of C-terminal mutations indicated in **a** are shown in green and blue. Right: GalS model with conserved and repeatedly mutated residue Arg146 colored cyan and the remaining mutations that occurred in the middle portion of the protein colored purple

Table S4). The frequency of and parallelism exhibited among hfq mutations is particularly striking in the context of experiments by Maharjan et al. where hfq mutations also arose, but remained at low frequency, being subject to negative frequency-dependent selection and epistatic interaction with rpoS mutations [38, 56, 69, 70].

Rho is a global regulator required for termination of nearly half of all *E. coli* transcripts. Like *hfq*, *rho* mutations can also be pleiotropic, resulting in either reduced or enhanced termination [83–88]. *rho* mutants have been recovered in *E. coli* populations adapted to high temperature [89, 90], ethanol stress [84, 85], and carbon source variation/limitation [29, 34, 91]. Mutagenesis and ChIP-chip analyses have shown that Rho-dependent terminators occur within genes that come under selection during glucose limitation, notably *lamB*, *mglA*, and *mglC* and downstream of *malT* and *mglC* [92, 93] and defective LamB expression in MalT activator mutants can be restored by compensatory mutations in *rho* [94].

Rho typically functions as a hexamer and termination requires binding of the RNA transcript as well as ATP hydrolysis. The primary RNA-binding domain is in the N-terminal half of the protein (aa 22-116) and the Cterminal half contains one ATP-hydrolysis domain (Ploop, aa 179-183) and two secondary RNA-binding domains (the Q-loop aa 278-290 and R-loop aa 322-326)



colored green and blue, respectively. Residues that are part of the MalK contact site but were not mutated are colored yellow. **c** Location and frequency of *malK* mutations on the primary structure. The N-terminal nucleotide-binding domain is colored white, and the C-terminal regulatory domain is shown in stipple. **d** Location of mutations on the 3D structure of a single MalK monomer. The C-terminal regulatory domain is colored light gray, and the N-terminal nucleotide-binding domain is colored dark gray. Observed nonsense mutations (blue, aa 339, 352), missense mutations observed here and reported to cause constitutive *mal* expression (purple, aa 267 and 297), missense mutations observed here but not reported elsewhere (cyan, aa 51, 225, 231, 253, 286, 296, 298, 349), and missense mutations reported to cause increased *mal* expression but not seen in this study (orange, aa 72, 248, 250, 251, 262, 268, 291, 346, 350) all occur in the same region of the C-terminal regulatory domain. **e** View of a MalK monomer with domains and mutations as in **b** rotated 180° along the *y*-axis

(Fig. 7) (as reviewed in [83]). Overall, we observed 11 mutations at 8 residues in Rho, 5 of which are associated with RNA-binding domains (aa 37, 87, 88, 278, and 293) and 4 of which interact with their neighboring subunit in the hexameric protein (aa 87, 88, 218, and 278) (Fig. 7a). Primary RNA-binding domain mutations at

residues 87-88 formed a cluster both in the amino acid sequence as well as the 3D structure (Table 1, Fig. 7). In addition, two high-frequency mutations (chemostat 1 Val278Phe, 93% by generation 100; chemostat 3 Ala293-Ser, 68% by generation 50) occurred on either side of the Q-loop secondary RNA-binding domain (Fig. 7c, d)



[95, 96]. We hypothesize that these high-frequency *rho* mutations enhance cells' capacity to scavenge limiting glucose by contributing to increased expression of both glycoporin LamB and inner membrane transporter MglBAC [92, 93, 97].

rho mutations in chemostats 1 and 3 fixed or nearly fixed early and did so in concert with mutations in MalK (chemostat 1 Ala296Asp) and mutations upstream of *mglB* (chemostats 1 and 3, bp 2,238,647) (Figs. 2 and 3; Additional file 2: Table S3 and Table S5; Additional file 5: Files S3; Additional file 7: File S5). By contrast, *rho* alleles detected in chemostat 2 never exceeded 6% frequency (Figs. 2, 3, and 7b; Additional file 2: Table S3).

Competition between sigma factors RpoD (σ^{D}) and RpoS (σ^{S}) for binding to core RNA polymerase directs transcription of genes sensitive to both sigma factors [98, 99]. Because many genes important for nutrient scavenging are downregulated when σ^{S} is abundant, loss-of-function mutations in *rpoS* are frequently favored during glucose-limited chemostat growth [100–102]. Our ancestral strain contains nonsense mutations in both *rpoS* (Gln33*) and *rpoD* (Glu26*) but with limited read-through of both transcripts enabled by the *sup*E44 suppressor [34].

rpoD transcript abundance is partly controlled by transcriptional termination at a highly conserved 31-nt rhoindependent terminator (T₁) between *rpsU* and *dnaG* (Additional file 1: Fig. S8) [103]. T₁ mutations predicted to decrease terminator stability were found early in chemostat 1 (bp 3,209,075 C-A, 92% frequency by generation 50) and later in chemostat 2 (bp 3,209,076 C→A and 3,209,082 G→T); one of these also occurred in the Helling et al. experiments (bp 3,209,075 C→A), suggesting it is beneficial under glucose limitation [36, 40]. We also cataloged a duplication in chemostat 2 that increases *rpoD* copy number. Although no duplications or SNPs in T₁ were observed in chemostat 3, in this chemostat, a lineage carrying an intragenic suppressor mutation in RpoD (*26Tyr) expanded to 4.7% of the population by generation 350. Overexpression of RpoD enabled by T₁ read-though, duplication, or intragenic suppression may ultimately increase transcription of operons positively controlled by σ^{70} (e.g., *mglBAC* and *malK-lamB-malM*), counteracting any residual σ^{S} stress response made possible by the supE44 suppressor.

Mutations that impact energy conservation, membrane biogenesis, and transcriptional run-through are laterarising targets of selection

Phosphoglucoseisomerase (Pgi)

Pgi is an abundantly expressed glycolytic enzyme that catalyzes the isomerization of hexose phosphates, thereby acting in both glycolysis and gluconeogenesis, and modulating pentose phosphate pathway (PPP) flux. While *pgi* is not essential in *E. coli, pgi* knockout mutants grow slowly on glucose, accumulate cAMP, reroute glucose-6-phosphate through the PPP, experience redox stress due to accumulation of NADPH, and utilize the glyoxylate shunt rather than the full TCA cycle [104–108]. Deletion of *pgi* has also been shown to favor

increased *mglBAC* and *lamB* transcription via CRPcAMP and targeted degradation of ptsG [104, 109]. In most species, including *E. coli*, functional Pgi exists as a dimer and mutation of residues across the interface between monomers has been hypothesized to alter subunit interactions and catalytic center geometry [110, 111].

Over the course of three replicate evolution experiments, we detected a total of 35 *pgi* alleles, 24 of which were unique (Additional file 2: Table S3, Fig. 8a. The large number of different variants suggests there must be some adaptive benefit to mutation of *pgi* and that the benefit is more likely due to reduced rather than enhanced Pgi function. When mapped onto the protein's 3D structure, many of the mutations localize to the dimerization interface (or just below it) and to the area near active site residues Glu355, His386, and Lys514 (Fig. 8b). Amino acid changes in either of these areas could be expected to inhibit Pgi activity.

Few *pgi* mutations rose to appreciable frequency before generation 200 (Fig. 3), suggesting their benefit may be contingent on other mutations or on some aspect of the chemostat environment that changed after this time point. *Pgi* alleles were least successful in chemostat 1, which was also the only replicate in which a large fraction of clones (79% by generation 500) acquired a second mutation upstream of *mglB*. This observation suggests that *pgi* mutations and mutations in the CRPbinding site of the *mglBAC* promoter may be functionally redundant.

Membrane glycosyltransferase

OPGH is involved in the synthesis of periplasmic glucans, highly branched oligosaccharides made from β -linked

Page 13 of 21

glucose monomers. While no opgH mutations are observed before generation 100, once they do appear, they rapidly increase in frequency, usually either just before or just after hfg mutations (Fig. 2, Additional file 1: Fig. S6, Additional file 5: Fig. S9, Additional file 6: Fig. S10, Additional file 7: Fig. S11). Novel opgH alleles, especially the nonsense mutations that we frequently observe (Additional file 1: Fig. S7), may constrain glucan production and serve as a glucose conservation measure. Also, a "moonlighting" function has recently been reported for OpgH: the glucosyltransferase interacts with the tubulinlike cell division protein FtsZ to delay cell division when levels of UDP-glucose are low [112]. Thus, OpgH mutations may augment the rate of cell division and thereby provide a fitness advantage under slow-growth chemostat conditions. The only opg operon mutation identified among strains in previous Adams et al. experiments occurred in opgG of the glucose scavenger, CV103 (E487*) [34]; we also observed 5 mutations in *opgG*.

Mutations that impact cell adhesion persist at low levels throughout our experiments *Fimbrial protein genes (fim)*

Genes associated with production/function of type 1 fimbriae, particularly *fimH* (fimbrial adhesion), were an unexpected and frequent target of mutation in all three chemostats (Table 1, Figs. 1 and 3; Additional file 2: Table S3; Additional file 5: Fig. S9, Additional file 6: Fig. S10, Additional file 7: Fig. S11). Though novel *fim* alleles were transient in vessels 2 and 3, in chemostat 1, a FimH Asn54Lys (corresponding to Asn33Lys in the mature protein) variant rose to a frequency of 70% by generation 150, temporarily displacing high-fitness alleles in *rho*,





malK, and upstream *mglB* (Additional file 5: Fig. S9). FimH mutants sometimes show an increased capacity for biofilm formation [113], a recurrent issue in chemostat experiments. However, as we failed to observe *fimH* mutant lineages acquire mutations expected to enhance glucose metabolism, the selection of *fimH* mutations here was more likely related to bacterial persistence than to competition for the limiting resource.

Discussion and conclusions

History matters: ancestry influences evolutionary trajectory

The evolutionary trajectory taken by a clonal population depends on its genetic point of departure. Our departure point was an ancestor carrying nonsense mutations in mismatch repair and in housekeeping and stationaryphase sigma factors, RpoD and RpoS, respectively. However, the ancestor also carried an amber/ochre/opal nonsense tRNA suppressor. Populations descending from such an ancestor carry an increased mutational load, but also have a built-in mechanism to lighten that load, specifically suppressing the burden of nonsense mutations that would otherwise result in loss-of-function (LOF). Laboratory evolution studies support the notion that LOF mutations can help drive adaptation [25, 114–116]. After all, metabolic networks can be reconfigured more easily by abolishing existing function(s) than by evolving altogether new ones [115]; thus, nonsense mutations or deletions sometimes confer greater fitness benefit than missense mutations affecting the same gene [116]. However, LOF may reduce metabolic flexibility by limiting the capacity of LOF mutants to compete in alternative environments [25].

Because our genetic "point of departure" was an rpoS nonsense mutant, it might be viewed as being preadapted to life under glucose limitation. After all, rpoS has repeatedly been shown to be a high-value target of selection under nutrient limitation (e.g., [38, 117, 118]); RpoS normally outcompetes housekeeping sigma factor RpoD for binding to RNA polymerase, repressing genes required for growth and cell division and activating those required to enter stationary phase [102, 119]. rpoS mutants with impaired function therefore continue to divide under conditions where wild-type cells arrest. However, the combined phenotypic effect of ancestral rpoS and rpoD nonsense mutations in a suppressor background is murky and begs the question: Is this particular combination of mutations favorable under glucose limitation, merely tolerated, or detrimental? Further, while many of the genetic changes we observed (e.g., those in galS, upstream mglB, hfq) enhance glucose assimilation, occur repeatedly, and go to high frequency, we also recovered clones that carry none of these mutations. Instead, these clones carry either intragenic suppressors of the ancestral *rpoD* nonsense mutation (*26 \rightarrow Asp and *26 \rightarrow Gln, chemostat 3 (Fig. 1)) or a duplication that increases *rpoD* copy number. Even when adaptation to one selective pressure is facilitated by LOF, and loss makes it difficult to adapt to another selective pressure, nonsense mutations have a distinct advantage over indels, because reversion or suppression of nonsense mutations is possible should environmental conditions change [25].

Another ancestral allele having the potential to influence evolutionary trajectory was the $A \rightarrow T$ CRP-binding site mutation 224 bp upstream of the low- K_m acetatescavenging enzyme, acs (acetyl-CoA synthetase). This mutation alters the regulation of the acs-pta operon such that the ancestor poorly assimilates low levels of the overflow metabolite acetate. In previous evolution experiments, this allele sometimes selectively favored semi-constitutive acs mutants capable of assimilating overflow acetate and attenuating its growth-inhibitory effects [35, 36, 39]. While we uncovered no evidence for this outcome, our failure to do so was not unanticipated: cross-feeding arose in only half the experiments founded by this ancestor or its close relatives [39]. A recent model [37] defining the boundary conditions for crossfeeding to evolve in a chemostat showed that the process is sensitive to variation in dilution rate as well as to the relative fitness of mutants that gain access to secondary metabolites. Subtle differences in either of these parameters may account for the apparent absence of the interaction. It is noteworthy that we detected no mutations at any of the loci previously implicated in cross-feeding evolution, e.g., acs, lpd, and ptsI [34, 120].

A large body of evidence points to acetyl-CoA synthetase being the route by which E. coli scavenge low levels of acetate from the extracellular environment [121, 122]. Indeed, to the best of our knowledge, this is the only route, though in principle a large decrease in the Km for acetate kinase (ack) might open another. However, we observed no mutations in ack. We also cannot rule out the possibility that multiple, different acs-mutant lineages co-exist, each below our detection limit, but together summing to >1%. However, we believe this is unlikely to have occurred in three independent replicates. Finally, we cannot exclude the possibility that a low-frequency acetate-scavenging contaminant arose and persisted in each of the three experiments, repeatedly escaping detection by our sensitive phage cocktail assay (see the "Methods" section).

A more plausible explanation lies in this: continuous glucose limitation selects not only for glucose scavenging, but also for efficiency of resource use [123, 124]. Because the latter places a premium on conserving the limiting carbon, we suggest that yet-to-be explored mutations among those we observed have the effect of restricting acetate overflow, which would confer a selective advantage. Consistent with this interpretation are observations that not all experimental populations which evolve glucose scavengers evolve acetate scavengers [39]. Adaptive lineages in the experiments reported here apparently found ways other than cross-feeding to consume all available carbon. As expected, throughout each evolution experiment, steady-state population size remained constant and residual glucose was at or below the limit of detection. Residual acetate (~ $45-90 \mu$ M) was observed early in each experiment (Additional file 1: Fig. S1), presumably owing to the ancestral acs regulatory mutation, but then fell below detection limit after generation 200. One possible mechanism for increased efficiency of glucose utilization lies in the proliferation of pgi mutants: generation 200 coincides with the emergence of novel pgi alleles in all three populations. Yao et al. reported that when pgi deletion mutants were grown in glucose-limited chemostats, glucose uptake rate dropped slightly compared to wild-type, but no overflow acetate was produced and biomass yield remained unchanged [125].

Population and clone sequencing open up a detailed view of the full spectrum of beneficial mutations and how that spectrum changes over time

High-coverage, whole-genome, whole-population sequencing makes it possible to discover every new allele reaching $\geq 1\%$ frequency in a population of $> 10^{10}$ cells. Because alleles are unlikely to reach such frequencies by drift, all were either transiently beneficial or hitchhiking with alleles that were. This depth of analysis opens up a richly detailed view of the spectrum of beneficial mutations arising in *E. coli* under constant resource limitation. Periodic whole-population sequencing allows patterns to be discerned as to how these spectra change over time, while clone sequencing makes it possible to establish linkage relations among novel alleles and represent their collective fate as evolving lineages.

Five general patterns emerge from these analyses. *First*, new alleles accumulate in replicate populations at similar rates, and the proportion of alleles that are missense, nonsense, synonymous, or noncoding remains fairly constant. *Second*, the distribution of new mutations across the genome is skewed with regard to their targets, with only a few dozen of the more than 1000 mutated genes being mutated more frequently than would be expected by chance alone; yet even among those most frequently mutated genes, few de novo mutations fix over the course of these experiments. *Third*, by clonal sequencing, we are able to establish that many, independent lineages co-exist and compete within the continuous cultures. Thus, evolutionary dynamics in these populations is governed by clonal interference and not by

clonal replacement or clonal reinforcement. This conclusion is reinforced by lack of evidence for mutations that support interactions which give cross-feeding consortia higher fitness and productivity than any consortium member by itself [120]. The dynamics of galS replacement illustrates the effect that clonal interference can have on the fate of different alleles. In chemostat 1, clones carrving GalS Arg146Leu rapidly dropped in frequency when lineages emerged with a mutation upstream of mglB (position 2,238,647); however, they were not completely displaced until generation 400 and even enjoyed brief periods of expansion. In chemostat 2, clones with the same mutation upstream of *mglB* were present by generation 50, but did not surpass a 90% threshold for another 250 generations due to competition from 22 different galS lineages and a lineage carrying a different upstream of mglB allele $(2,238,648 \text{ G} \rightarrow \text{T})$ (Figs. 2 and 3b, Additional file 4: Table S7). By contrast, in chemostat 3, a lineage with the upstream *mglB* mutation (2,238,647) experienced little competition and was nearly fixed by generation 150 (Additional file 2: Table S5).

A *fourth* pattern to emerge is widespread parallelism in regulatory evolution. Both across and within populations, the same genes are mutated again and again, often at exactly the same nucleotide position in independent replicates, and sometimes in independent lineages coevolving within the same vessel. Many of these genes (galS, malT, malK, upstream mglB, hfq, rho) act in processes related to the transport and assimilation of the limiting nutrient, glucose. However, in most cases, the mutations recovered alter regulation of these processes, and not the structural proteins that carry them out. This finding is consistent with a number of recent experimental evolution studies using E. coli, where the genetic basis for adaptive change could be traced back to mutations in regulatory elements or regulatory loci, e.g., [126–129] and references therein.

A *fifth* pattern relates to the order of beneficial mutations and the influence that order has on evolutionary dynamics. Consistent with previous reports, mutations that increase glucose flux across the inner membrane (galS, upstream mglB) occur early and precede those that increase flux across the outer membrane (malK/ *malT*, *hfq*, *rho*). In both cases, mutations in binding partners (GalS/upstream mglB and MalT/MalK) rarely occur in the same clone, and the order in which they occur can lead to either a sweep (upstream mglB clones quickly displace galS clones) or to clonal interference (malT and malK clones can co-exist). Other alleles emerge later and nearly always together: clones with existing mutations in the mal operon acquire subsequent mutations in hfq and opgH, regardless of which gene is altered first or which alleles are already present in the population. These patterns are reminiscent of genotypes described by Kinnersley et al. [22] in which glucose scavenger CV103 has mutations in *malK*, *opgG*, and *hfq* while acetate specialist CV101 only carries a mutation in *malT*.

With regard to periodic selection, rather than favorable alleles arising within a set of lineages that successively replace one another over time, we observe groups or cohorts of mutations co-evolving, with widespread clonal interference among lineages that carry different beneficial mutations [130]. For example, in chemostat 1, a spreading lineage with a cohort of mutations upstream of *mglB/lptA/opgH* (pink) is impeded by the emergence of lineages carrying mutations in hfq (green) (Additional file 5: Fig. S9). All of these phenomena-hard and soft sweeps, cohorts of mutations that increase or decrease in frequency together, and clonal interference-have been observed in yeast [18, 24, 131] and E. coli [29] populations evolving in the laboratory, as well as in Pseudomonas aeruginosa evolving in the cystic fibrosis lung [132].

Similar experiments carried out by Maharjan et al. [38] using *E. coli* BW2952 showed that population-level phenotypic changes in glucose-limited chemostats are often the result of multiple soft sweeps by combinations of beneficial mutations. While we did not assay clone phenotypes, multiple alleles of galS, hfq, and opgH appear to sweep our populations in concert suggesting a similar pattern in which a phenotypic effect (reduced expression of a particular gene) is favored, but has different genetic bases in co-existing lineages. At the clone level, BW2952 exhibits sign epistasis between mutations in rpoS/hfq and galS/malT [38, 56]. In our experiments, we found no evidence of sign epistasis between the ancestral rpoS allele and hfq: by generation 250, over 50% of clones in populations 1 and 3 carry mutations in both genes. Maharjan et al. have suggested that fitness deficits in *rpoS/hfq* double mutants may arise from altered cell division [69, 133], specifically, hfq mutations that enhance glucose uptake during slow growth, may diminish viability when cells divide rapidly. Hfq deletion mutants exhibit cell division anomalies due to elevated expression of cell division proteins, including FtsZ [134, 135]. Interestingly, during fast growth, OpgH (which in our experiments is nearly always mutated alongside hfq binds FtsZ to postpone cell division [112]. Thus, it may be that in our experiments the negative fitness effects associated with hfq-rpoS double mutants are mitigated by mutations in opgH. We should also note that cells in the Maharjan et al. evolution experiments were subject to a dilution rate of $D = 0.1 \text{ h}^{-1}$, whereas those in experiments performed by Adams et al. were dividing twice as fast $(D = 0.2 \text{ h}^{-1})$. Thus, this discrepancy may be a manifestation of trade-offs between glucose uptake and cell viability. Finally, some mutations occur repeatedly and are therefore likely adaptive, yet their dynamics are unpredictable: for example, beneficial mutations in transcriptional terminator *rho* sweep when they co-occur with beneficial mutations upstream of *mglB*, but otherwise remain at low frequency (Fig. 3, Additional file 2: Table S3). This dependence on genetic context, or "quasi-hitchhiking," of beneficial mutations was previously observed by Lang et al. in yeast and may be a feature that only becomes evident when experimental populations are sequenced to high depth of coverage and at sufficient temporal resolution [18].

Classic studies by Adams and colleagues showed that populations originating from the same ancestor used in our experiments could evolve into consortia consisting of ecotypes that co-exist via cross-feeding [35, 36] and relief from product inhibition [37]. Recent data have shown that such consortia can be more fit and are more productive than either their ancestor or clones representing each ecotype [120]. By contrast, in the populations described here, neither the observed spectrum of mutations nor the structure of clone phylogenies suggest trophic interactions that could be construed as clonal reinforcement. This finding is consistent with Treves et al. [39] who found that cross-feeding arose in only 6 of 12 replicate E. coli populations evolved under conditions identical to those we used and founded by closely related ancestors. In each of these populations, acetate scavenging was driven by regulatory mutations that resulted in semi-constitutive overexpression of acetyl-CoA synthetase (acs) [39]. These mutants were supported by glucose-scavenging small colony variants that were evident on plates between generations 88 and 381, depending on replicate. In our experiments, acs mutants never arose, or if they did, their frequency never exceeded our 1% level of detection. Likewise, mutations thought to contribute to the acetate excretion phenotype [34] were not observed either in our population sequencing or clone sequencing data. Thus, as was the case for half the populations examined by Treves et al. [39], the molecular mechanisms to support acetate cross-feeding were not established over the time course of our experiments.

Methods

Strains, media, and culture conditions

Escherichia coli JA122, population samples and clones were maintained as permanent frozen stocks and stored at -80 °C in 20% glycerol. Davis minimal medium was used for all liquid cultures with 0.025% glucose added for batch cultures and 0.0125% for chemostats, as previously described [40]. Chemostat cultures were initiated using independent colonies picked from a Tryptone Agar (TA) plate inoculated with JA122, then outgrown in Davis minimal batch medium overnight. Consistent with previous evolution experiments founded by this

ancestor, chemostats were maintained at 30 °C with a dilution rate of $\approx 0.2 \,h^{-1}$ for 300–500 generations. Every other day, population samples were archived in duplicate at – 80 °C; culture density and purity were assessed by measuring absorbance at A₅₅₀ and by plating serial dilutions on TA and examining colony-forming units (CFU) following 24-h incubation at 30 °C. When necessary, chemostats were re-started from frozen stocks (chemostat 1: generation 217; chemostat 2: generation 410; chemostat 3 generation 251). At each sequencing time point, 50 mL of culture was pelleted then stored at – 80 °C for DNA extraction. For clone sequencing, entire colonies were picked from TA plates inoculated from glycerol stocks and re-archived in 96-well plate format.

Phage cocktail assay of culture purity

To assay chemostat cultures for contamination, lysates of bacteriophages T2, T5, and a T6 mutant were prepared using the ancestral strain JA122 as a host, following procedures first described by [41]. Lysates were filtered through a 0.2- μ M filter to remove cell debris and concentrated with a 10-kDa MWCO filter to bring the concentration of each phage to $2.5 \times 10^{12} \text{ mL}^{-1}$. Every 10–20 generations, 200 μ L of phage cocktail was applied to the surface of a TA agar plate, dried, and used to screen 100 μ L of chemostat culture for non-*E. coli* contaminants.

Metabolite assays

Ten milliliters of sterile, cell-free chemostat filtrate was concentrated 20-fold by lyophilization (Labconco 4.5 Liter Freeze Dry System), then re-suspended in 0.5 mL sterile Millipore water. Residual glucose and residual acetate concentrations were determined on concentrated filtrate. Glucose was assayed enzymatically using the High Sensitivity Glucose Assay Kit (Sigma-Aldrich, Cat# MAK181), while acetate concentration was determined using the Acetate Colorimetric Assay Kit (Sigma-Aldrich, Cat# MAK086). Results presented in Additional file 1: Fig. S1 represent means ± SEM of duplicate assays.

Population sequencing

Bacterial DNA was prepared using the DNeasy Blood and Tissue Kit (Qiagen, cat. 69504) following the manufacturer's guidelines. For population sequencing, 5×10^{10} cells, collected from every 50 generations in three chemostat vessels (up to 500 generations in vessels 1 and 2, and up to 300 generations in vessel 3, 29 samples total) and frozen as pellets, yielded 10–20 µg of DNA. Following Proteinase K treatment, RNaseA treatment was used (20 µL 10 mg/mL RNAse A, 2 min at room temperature) to avoid degraded RNA from visually obscuring size selection during library preparation. Samples were split into two columns to avoid overloading. Bacterial DNA was sheared to a 150–200-bp fragment size using a Covaris S2 series sonicator (6 min, duty = 5%, intensity = 3, cycles/burst = 200) and was then ligated to barcoded adapters as described [136], except that 200-bp fragments were size selected after adapter ligation (to maximize the fidelity of sequencing, by reading each fragment in both directions). Six barcoded libraries were combined and sequenced on each lane of HiSeq 2000 Illumina Sequencer.

Variant calling from population sequencing with CLC Genomics Workbench 7.5

Illumina reads were trimmed (removing adapters on both ends) and stringently mapped (mismatch cost 2, insertion cost 3, deletion cost 3, length fraction 1.0, similarity fraction 0.97) to the reference sequence (WIS_ MG1655_m56). Variants were called with the following parameters: minimum frequency 1%, minimal coverage 100, minimum count 2, and base quality filtering (neighborhood radius 5, minimum central quality 15, and minimum neighborhood quality 20). Sequencing data uncovered low-level contamination of whole population samples with Serratia liquifaciensis. We therefore first determined the proportion of contaminating reads by mapping population sequencing to S. liquifaciensis genome and then removed SNPs with frequency closely tracking the percentage of contamination (between 1 and 5%) that matched *S. liquifaciensis* sequence.

Selection of clones for sequencing

Allele frequencies for each chemostat were examined at each time point, and the time point at which there was the largest number of alleles present at 5% or greater frequency was chosen for the isolation of clones for whole-genome sequencing. The rationale for this was that it would afford us the greatest opportunity to phase as many high-frequency alleles as possible.

Clonal DNA preparation

A colony was re-suspended in $300 \,\mu\text{L}$ of sterile ddH₂0 with 17% glycerol and stored in three aliquots at – $80 \,^{\circ}\text{C}$. One hundred microliters of glycerol stock was used for DNA preparation. After removing glycerol (using Multi-Screen High Volume Filter Plates with 0.45 μ m Durapore membrane, Millipore MVHVN4525), cells were resuspended in 500 μ L LB and grown overnight at 30 °C without shaking in deep well plates. Cells were collected again using filter plates and subjected to DNeasy 96 Blood and Tissue Kit (Qiagen 69581) (yielding 4–15 μ g per strain).

Clonal library preparation and sequencing

Multiplexed sequencing libraries from clones were prepared using the Nextera kit (Illumina catalog # FC-121-1031 and # FC-121-1012) as described in [137], starting with 1–4 ng of genomic DNA. Resulting libraries from each 96-well plate were pooled at an equal volume. Resulting pooled libraries were analyzed on the Qubit and Bioanalyzer platforms and sequenced on HiSeq 2000 (one lane per 96 clone pool). All raw sequencing data are available from the SRA under BioProject ID PRJNA517527.

Variant calling from clonal sequencing with CLC Genomics Workbench 7.5

Short reads with adapters removed were mapped to the reference with the same parameters as above, except the length fraction was set to 0.5, and the similarity fraction to 0.8. Variants were called with a minimum frequency 80%, minimum count 2, and the same base quality filtering as above.

Generation of phylogenies

For each chemostat, SNP and indel events for all 96 clones and the ancestor JA122 were concatenated and re-coded as binary characters (i.e., presence/absence with the ancestral state composed of all zeroes) and assembled into character matrices. PAUP ver. 4.0a149 was used to generate Camin-Sokal parsimony trees using the ancestor as the outgroup under the assumption that reversions were extremely unlikely due to the extreme transversion bias [138, 139]. Tree files (.tre) were loaded into the Interactive Tree of Life (iTOL) web service for character mapping and figure generation [140].

Determining genes with an excess of mutations

To identify genes with an excess of mutations, we first determined the overall density of mutations as:

 $\rho = M/L$, where M is the total number of mutations and *L* is the length of the genome.

The probability of a given mutation landing in a segment of length l is:

 $\lambda = \rho \times l$

To calculate the p value of n mutations landing in a segment of length l, we assume a Poisson sampling process of a mutation landing in a given segment and thus use:

$$p = \sum_{i=n}^{\infty} \frac{\lambda^i \ge e^{-\lambda}}{i!}$$

though, in practice, we capped *i* arbitrarily at 50, as continually summing at i > 50 does not appreciably affect the calculated *p* value. For a given segment, we calculated the number of segments that would be expected to have *p* value as good or better, as the number of tested segments multiplied by the *p* value. From this, we also determined a false positive rate.

Generation of Muller diagrams

Based on both the clonal sequencing, we were able to determine which mutations were in which lineages together, and from both the clonal and population sequencing an approximate order of those mutations (though this was not exhaustive for all mutations). Using these data, we developed a lineage file format that described which mutations occurred in which lineages, and which lineages descended from one another, and used a custom Perl script that combined this information with the allele frequencies over time from the population sequencing to generate a graphical representation of the evolutionary dynamics, often referred to as a Muller diagram.

Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s12915-021-00954-0.

Additional file 1 Fig. S1: Cell density and residual metabolite concentrations. Fig. S2: Input of de novo mutations. Fig. S3: Most de novo mutations only reach low allele frequencies and experience pervasive clonal interference. (A) Histogram of maximum frequencies; (B) Final vs. maximum frequencies; (C) Venn diagram showing degree of genic parallelism among beneficial mutations. Fig. S4: Isolated clones are representative of the populations from which they are drawn. Fig. S5: MalK/MalT population dynamics. Fig. S6: Mutations in glucosyltransferase opgH occur repeatedly and, collectively, go to high frequency. Fig. S7: opgH has nonsense and missense mutations throughout its length. Fig. S8: Mutations that decrease T1 terminator stability in the macromolecular synthesis operon are expected to affect expression of dnaG (DNA primase) and rpoD (housekeeping σ-factor). Predicted Δ G values for wild-type T1 terminator from *E. coli* K12 MG1655 and two variants observed in chemostat 1 were determined using unafold.rna.albany.edu/?q=mfold/RNA-Folding-Form. The C-A mutation at nucleotide 3,209,075 has been previously observed in chemostatevolved E. coli [34].

Additional file 2 Table S1. Key mutations that distinguish ancestral strain JA122 from K12 (MG1655). Table S2. Beneficial alleles. Table S3. Population allele frequencies for frequently mutated genes. Table S4. Identical mutations arise within and among replicate evolution experiments. Table S5. Fixed alleles among replicate populations ("fixed" defined as > 98% at any time point between generation 50 and 500).

Additional file 3 Table S6. Spreadsheet containing identity and frequencies of all mutations detected via population sequencing.

Additional file 4 Table 57. Alleles mapped onto clone phylogenies represented in main Fig. 1.

Additional file 5 Fig. S9. Muller diagrams for novel alleles arising in chemostat 1, showing details for each lineage.

Additional file 6 Fig. S10. Muller diagrams for novel alleles arising in chemostat 2, showing details for each lineage.

Additional file 7 Fig. S11. Muller diagrams for novel alleles arising in chemostat 3, showing details for each lineage.

Additional file 8 Fig. S12. Locations of mutations in genes that were targets of adaptation, and their maximum frequencies, on both log and linear scales.

Acknowledgements

The authors thank Emily Cook, Matthew Herron, Jacob Boswell, and Eugene Kroll for their careful reading of the manuscript and thoughtful suggestions for its improvement.

Authors' contributions

Conceived and designed the experiments: GS, MK, and FR. Performed the experiments: MK and KS. Analyzed the data: GS, KS, MK, DY, and FR. Contributed reagents/materials/analysis tools: GS. Wrote the paper: MK, KS, GS, and FR. All authors read and approved the final manuscript.

Funding

This work was funded by NASA grants NNX12AD87G, 80NSSC20K0621, and NNA17BB05A (Astrobiology Institute) to FR and GS and by NIH grant R35 GM131824 to GS.

Availability of data and materials

All raw sequencing data are available from the SRA under BioProject ID PRJNA517527 [141].

Ethics approval and consent to participate

Not applicable

Consent for publication

Not applicable

Competing interests

The authors declare that they have no competing interests.

Author details

¹Division of Biological Sciences, The University of Montana, Missoula, MT 59812, USA. ²Department of Genetics, Stanford University School of Medicine, Stanford, CA 94305-5120, USA. ³School of Biological Sciences, Georgia Institute of Technology, Atlanta, GA 30332, USA.

Received: 2 July 2020 Accepted: 8 January 2021 Published online: 04 February 2021

References

- 1. Lang Gl, Desai MM. The spectrum of adaptive mutations in experimental evolution. Genomics. 2014;104:412–6.
- Cvijovic I, Nguyen Ba AN, Desai MM. Experimental studies of evolutionary dynamics in microbes. Trends Genet. 2018;34:693–703.
- Cooper VS: Experimental evolution as a high-throughput screen for genetic adaptations. mSphere 2018;3(3):e00121-18.
- Garay E, Campos SE, Gonzalez de la Cruz J, Gaspar AP, Jinich A, Deluna A: High-resolution profiling of stationary-phase survival reveals yeast longevity factors and their genetic interactions. Plos Genet 2014, 10:e1004168.
- Paradis-Bleau C, Kritikos G, Orlova K, Typas A, Bernhardt TG. A genome-wide screen for bacterial envelope biogenesis mutants identifies a novel factor involved in cell wall precursor metabolism. PLoS Genet. 2014;10:e1004056.
- Novick A, Szilard L. Experiments with the chemostat on spontaneous mutations of bacteria. Proc Natl Acad Sci U S A. 1950;36:708–19.
- Novick A, Szilard L. Experiments on spontaneous and chemically induced mutations of bacteria growing in the chemostat. Cold Spring Harb Symp Quant Biol. 1951;16:337–43.
- Atwood KC, Schneider LK, Ryan FJ. Periodic selection in Escherichia coli. Proc Natl Acad Sci U S A. 1951;37:146–55.
- Atwood KC, Schneider LK, Ryan FJ. Selective mechanisms in bacteria. Cold Spring Harb Symp Quant Biol. 1951;16:345–55.
- Haldane JBS. A mathematical theory of natural and artificial selection, part V: selection and mutation. Math Proc Camb Philos Soc. 1927;23:838–44.
- 11. Fisher RAS. The genetical theory of natural selection. Oxford: Clarendon Press; 1930.
- 12. Muller HJ. Some genetic aspects of sex. Am Nat. 1932;66:118-38.
- Gause GF. Experimental analysis of Vito Volterra's mathematical theory of the struggle for existence. Science. 1934;79:16–7.
- Liang Y, Yin X, Zeng L, Chen S. Clonal replacement of epidemic KPCproducing Klebsiella pneumoniae in a hospital in China. BMC Infect Dis. 2017;17:363.
- Knol MJ, Hahne SJM, Lucidarme J, Campbell H, de Melker HE, Gray SJ, Borrow R, Ladhani SN, Ramsay ME, van der Ende A. Temporal associations between national outbreaks of meningococcal serogroup W and C disease in the Netherlands and England: an observational cohort study. Lancet Public Health. 2017;2:e473–82.

- Caswell-Jin JL, McNamara K, Reiter JG, Sun R, Hu Z, Ma Z, Ding J, Suarez CJ, Tilk S, Raghavendra A, et al. Clonal replacement and heterogeneity in breast tumors treated with neoadjuvant HER2-targeted therapy. Nat Commun. 2019;10:657.
- Yost KE, Satpathy AT, Wells DK, Qi Y, Wang C, Kageyama R, McNamara KL, Granja JM, Sarin KY, Brown RA, et al. Clonal replacement of tumor-specific T cells following PD-1 blockade. Nat Med. 2019;25:1251–9.
- Lang GI, Rice DP, Hickman MJ, Sodergren E, Weinstock GM, Botstein D, Desai MM. Pervasive genetic hitchhiking and clonal interference in forty evolving yeast populations. Nature. 2013;500:571–4.
- 19. Marad DA, Buskirk SW, Lang GI. Altered access to beneficial mutations slows adaptation and biases fixed mutations in diploids. Nat Ecol Evol. 2018;2:882–9.
- Levy SF, Blundell JR, Venkataram S, Petrov DA, Fisher DS, Sherlock G. Quantitative evolutionary dynamics using high-resolution lineage tracking. Nature. 2015;519:181–6.
- Good BH, McDonald MJ, Barrick JE, Lenski RE, Desai MM. The dynamics of molecular evolution over 60,000 generations. Nature. 2017;551:45–50.
- Lauer S, Avecilla G, Spealman P, Sethia G, Brandt N, Levy SF, Gresham D. Single-cell copy number variant detection reveals the dynamics and diversity of adaptation. PLoS Biol. 2018;16:e3000069.
- Blundell JR, Schwartz K, Francois D, Fisher DS, Sherlock G, Levy SF. The dynamics of adaptive genetic diversity during the early stages of clonal evolution. Nat Ecol Evol. 2019;3:293–301.
- 24. Kao KC, Sherlock G. Molecular characterization of clonal interference during adaptive evolution in asexual populations of Saccharomyces cerevisiae. Nat Genet. 2008;40:1499–504.
- 25. Kvitek DJ, Sherlock G. Whole genome, whole population sequencing reveals that loss of signaling networks is the major adaptive strategy in a constant environment. PLoS Genet. 2013;9:e1003972.
- Evans JA, McDonald SA. The complex, clonal, and controversial nature of Barrett's esophagus. Adv Exp Med Biol. 2016;908:27–40.
- Behringer MG, Choi BI, Miller SF, Doak TG, Karty JA, Guo W, Lynch M. Escherichia coli cultures maintain stable subpopulation structure during long-term evolution. Proc Natl Acad Sci U S A. 2018;115:E4642–50.
- Rozen DE, Lenski RE. Long-term experimental evolution in Escherichia coli. VIII. Dynamics of a balanced polymorphism. Am Nat. 2000;155:24–35.
- Herron MD, Doebeli M. Parallel evolutionary dynamics of adaptive diversification in Escherichia coli. PLoS Biol. 2013;11:e1001490.
- Maddamsetti R, Lenski RE, Barrick JE. Adaptation, clonal interference, and frequency-dependent interactions in a long-term evolution experiment with Escherichia coli. Genetics. 2015;200:619–31.
- 31. Charlesworth D. Balancing selection and its effects on sequences in nearby genome regions. PLoS Genet. 2006;2:e64.
- 32. Rainey PB, Travisano M. Adaptive radiation in a heterogeneous environment. Nature. 1998;394:69–72.
- Rozen DE, Schneider D, Lenski RE. Long-term experimental evolution in Escherichia coli. XIII. Phylogenetic history of a balanced polymorphism. J Mol Evol. 2005;61:171–80.
- Kinnersley M, Wenger J, Kroll E, Adams J, Sherlock G, Rosenzweig F. Ex uno plures: clonal reinforcement drives evolution of a simple microbial community. PLoS Genet. 2014;10:e1004430.
- Rosenzweig RF, Sharp RR, Treves DS, Adams J. Microbial evolution in a simple unstructured environment: genetic differentiation in Escherichia coli. Genetics. 1994;137:903–17.
- 36. Helling RB, Vargas CN, Adams J. Evolution of Escherichia coli during growth in a constant environment. Genetics. 1987;116:349–58.
- Gudelj I, Kinnersley M, Rashkov P, Schmidt K, Rosenzweig F. Stability of cross-feeding polymorphisms in microbial communities. PLoS Comput Biol. 2016;12:e1005269.
- Maharjan RP, Liu B, Feng L, Ferenci T, Wang L. Simple phenotypic sweeps hide complex genetic changes in populations. Genome Biol Evol. 2015;7:531–44.
- Treves DS, Manning S, Adams J. Repeated evolution of an acetatecrossfeeding polymorphism in long-term populations of Escherichia coli. Mol Biol Evol. 1998;15:789–97.
- Kinnersley MA, Holben WE, Rosenzweig F. E Unibus Plurum: genomic analysis of an experimentally evolved polymorphism in Escherichia coli. PLoS Genet. 2009;5:e1000713.
- Lenski RE, Rose MR, Simpson SC, Tadler SC. Long-term experimental evolution in Escherichia coli. I. Adaptation and divergence during 2,000 generations. Am Nat. 1991;138:1315–41.
- Atlung T, Nielsen HV, Hansen FG. Characterisation of the allelic variation in the rpoS gene in thirteen K12 and six other non-pathogenic Escherichia coli strains. Mol Gen Genomics. 2002;266:873–81.

- Singaravelan B, Roshini BR, Munavar MH. Evidence that the supE44 mutation of Escherichia coli is an amber suppressor allele of glnX and that it also suppresses ochre and opal nonsense mutations. J Bacteriol. 2010;192:6039–44.
- Au KG, Clark S, Miller JH, Modrich P. Escherichia coli mutY gene encodes an adenine glycosylase active on G-A mispairs. Proc Natl Acad Sci U S A. 1989; 86:8877–81.
- Blattner FR, Plunkett G 3rd, Bloch CA, Perna NT, Burland V, Riley M, Collado-Vides J, Glasner JD, Rode CK, Mayhew GF, et al. The complete genome sequence of Escherichia coli K-12. Science. 1997;277:1453–62.
- Lee H, Popodi E, Tang H, Foster PL. Rate and molecular spectrum of spontaneous mutations in the bacterium Escherichia coli as determined by whole-genome sequencing. Proc Natl Acad Sci U S A. 2012;109:E2774–83.
- de Visser JA, Rozen DE. Clonal interference and the periodic selection of new beneficial mutations in Escherichia coli. Genetics. 2006;172:2093–100.
- Hermisson J, Pennings PS. Soft sweeps: molecular population genetics of adaptation from standing genetic variation. Genetics. 2005;169:2335–52.
- Pennings PS, Hermisson J. Soft sweeps II–molecular population genetics of adaptation from recurrent mutation or migration. Mol Biol Evol. 2006;23:1076–84.
- Pennings PS, Hermisson J. Soft sweeps III: the signature of positive selection from recurrent mutation. PLoS Genet. 2006;2:e186.
- Desai MM, Walczak AM, Fisher DS. Genetic diversity and the structure of genealogies in rapidly adapting populations. Genetics. 2013;193:565–85.
- Notley-McRobb L, Ferenci T. The generation of multiple co-existing malregulatory mutations through polygenic evolution in glucose-limited populations of Escherichia coli. Environ Microbiol. 1999;1:45–52.
- Hermisson J, Pennings PS. Soft sweeps and beyond: understanding the patterns and probabilities of selection footprints under rapid adaptation. Methods Ecol Evol. 2017;8:700–16.
- 54. Jensen JD. On the unfounded enthusiasm for soft selective sweeps. Nat Commun. 2014;5:5281.
- 55. Ferenci T. Hungry bacteria–definition and properties of a nutritional state. Environ Microbiol. 2001;3:605–11.
- Maharjan RP, Ferenci T. Epistatic interactions determine the mutational pathways and coexistence of lineages in clonal Escherichia coli populations. Evolution. 2013;67:2762–8.
- Manch K, Notley-McRobb L, Ferenci T. Mutational adaptation of Escherichia coli to glucose limitation involves distinct evolutionary pathways in aerobic and oxygen-limited environments. Genetics. 1999;153:5–12.
- Notley-McRobb L, Ferenci T. Adaptive mgl-regulatory mutations and genetic diversity evolving in glucose-limited Escherichia coli populations. Environ Microbiol. 1999;1:33–43.
- Notley-McRobb L, Ferenci T. Experimental analysis of molecular events during mutational periodic selections in bacterial evolution. Genetics. 2000;156:1493–501.
- Notley-McRobb L, Seeto S, Ferenci T. The influence of cellular physiology on the initiation of mutational pathways in Escherichia coli populations. Proc Biol Sci. 2003;270:843–8.
- Ferenci T. Adaptation to life at micromolar nutrient levels: the regulation of Escherichia coli glucose transport by endoinduction and cAMP. FEMS Microbiol Rev. 1996;18:301–17.
- Death A, Ferenci T. The importance of the binding-protein-dependent Mgl system to the transport of glucose in Escherichia coli growing on low sugar concentrations. Res Microbiol. 1993;144:529–37.
- Death A, Notley L, Ferenci T. Derepression of LamB protein facilitates outer membrane permeation of carbohydrates into Escherichia coli under conditions of nutrient stress. J Bacteriol. 1993;175:1475–83.
- 64. Geanacopoulos M, Adhya S. Functional characterization of roles of GaIR and GaIS as regulators of the gal regulon. J Bacteriol. 1997;179:228–34.
- Hollands K, Busby SJ, Lloyd GS. New targets for the cyclic AMP receptor protein in the Escherichia coli K-12 genome. FEMS Microbiol Lett. 2007;274:89–94.
- Weickert MJ, Adhya S. The galactose regulon of Escherichia coli. Mol Microbiol. 1993;10:245–51.
- Schumacher MA, Glasfeld A, Zalkin H, Brennan RG. The X-ray structure of the PurR-guanine-purF operator complex reveals the contributions of complementary electrostatic surfaces and a water-mediated hydrogen bond to corepressor specificity and binding affinity. J Biol Chem. 1997;272:22648–53.
- Fukami-Kobayashi K, Tateno Y, Nishikawa K. Parallel evolution of ligand specificity between Lacl/GalR family repressors and periplasmic sugarbinding proteins. Mol Biol Evol. 2003;20:267–77.
- Maharjan R, McKenzie C, Yeung A, Ferenci T. The basis of antagonistic pleiotropy in hfq mutations that have opposite effects on fitness at slow and fast growth rates. Heredity (Edinb). 2013;110:10–8.

- Maharjan R, Zhou Z, Ren Y, Li Y, Gaffe J, Schneider D, McKenzie C, Reeves PR, Ferenci T, Wang L. Genomic identification of a novel mutation in hfq that provides multiple benefits in evolving glucose-limited populations of Escherichia coli. J Bacteriol. 2010;192:4517–21.
- Richet E, Joly N, Danot O. Two domains of MalT, the activator of the Escherichia coli maltose regulon, bear determinants essential for antiactivation by MalK. J Mol Biol. 2005;347:1–10.
- Dardonville B, Raibaud O. Characterization of malT mutants that constitutively activate the maltose regulon of Escherichia coli. J Bacteriol. 1990;172:1846–52.
- Bohm A, Diez J, Diederichs K, Welte W, Boos W. Structural model of MalK, the ABC subunit of the maltose transporter of Escherichia coli: implications for mal gene regulation, inducer exclusion, and subunit assembly. J Biol Chem. 2002;277:3708–17.
- Kuhnau S, Reyes M, Sievertsen A, Shuman HA, Boos W. The activities of the Escherichia coli MalK protein in maltose transport, regulation, and inducer exclusion can be separated by mutations. J Bacteriol. 1991;173:2180–6.
- Schlegel A, Danot O, Richet E, Ferenci T, Boos W. The N terminus of the Escherichia coli transcription activator MaIT is the domain of interaction with MaIY. J Bacteriol. 2002;184:3069–77.
- Danot O, Raibaud O. Multiple protein-DNA and protein-protein interactions are involved in transcriptional activation by MaIT. Mol Microbiol. 1994;14: 335–46.
- Vidal-Ingigliardi D, Richet E, Danot O, Raibaud O. A small C-terminal region of the Escherichia coli MaIT protein contains the DNA-binding domain. J Biol Chem. 1993;268:24527–30.
- Lippincott J, Traxler B. MalFGK complex assembly and transport and regulatory characteristics of MalK insertion mutants. J Bacteriol. 1997;179:1337–43.
- Kvitek DJ, Sherlock G. Reciprocal sign epistasis between frequently experimentally evolved adaptive mutations causes a rugged fitness landscape. PLoS Genet. 2011;7:e1002056.
- Guisbert E, Rhodius VA, Ahuja N, Witkin E, Gross CA. Hfq modulates the sigmaE-mediated envelope stress response and the sigma32-mediated cytoplasmic stress response in Escherichia coli. J Bacteriol. 2007;189:1963–73.
- Moller P, Overloper A, Forstner KU, Wen TN, Sharma CM, Lai EM, Narberhaus F. Profound impact of Hfq on nutrient acquisition, metabolism and motility in the plant pathogen Agrobacterium tumefaciens. PLoS One. 2014;9: e110427.
- Maharjan RP, Ferenci T, Reeves PR, Li Y, Liu B, Wang L. The multiplicity of divergence mechanisms in a single evolving population. Genome Biol. 2012; 13:R41.
- 83. Mitra P, Ghosh G, Hafeezunnisa M, Sen R. Rho protein: roles and mechanisms. Annu Rev Microbiol. 2017;71:687–709.
- Haft RJ, Keating DH, Schwaegler T, Schwalbach MS, Vinokur J, Tremaine M, Peters JM, Kotlajich MV, Pohlmann EL, Ong IM, et al. Correcting direct effects of ethanol on translation and transcription machinery confers ethanol tolerance in bacteria. Proc Natl Acad Sci U S A. 2014;111:E2576–85.
- Freddolino PL, Goodarzi H, Tavazoie S. Fitness landscape transformation through a single amino acid change in the rho terminator. PLoS Genet. 2012;8:e1002744.
- Cardinale CJ, Washburn RS, Tadigotla VR, Brown LM, Gottesman ME, Nudler E. Termination factor Rho and its cofactors NusA and NusG silence foreign DNA in E. coli. Science. 2008;320:935–8.
- 87. Banerjee S, Chalissery J, Bandey I, Sen R. Rho-dependent transcription termination: more questions than answers. J Microbiol. 2006;44:11–22.
- Martinez A, Opperman T, Richardson JP. Mutational analysis and secondary structure model of the RNP1-like sequence motif of transcription termination factor Rho. J Mol Biol. 1996;257:895–908.
- Gonzalez-Gonzalez A, Hug SM, Rodriguez-Verdugo A, Patel JS, Gaut BS. Adaptive mutations in RNA polymerase and the transcriptional terminator rho have similar effects on Escherichia coli gene expression. Mol Biol Evol. 2017;34:2839–55.
- Tenaillon O, Rodriguez-Verdugo A, Gaut RL, McDonald P, Bennett AF, Long AD, Gaut BS. The molecular diversity of adaptive convergence. Science. 2012;335:457–61.
- Le Gac M, Cooper TF, Cruveiller S, Medigue C, Schneider D. Evolutionary history and genetic parallelism affect correlated responses to evolution. Mol Ecol. 2013;22:3292–303.
- Peters JM, Mooney RA, Kuan PF, Rowland JL, Keles S, Landick R. Rho directs widespread termination of intragenic and stable RNA transcription. Proc Natl Acad Sci U S A. 2009;106:15406–11.

- 93. Ciampi MS. Rho-dependent terminators and transcription termination. Microbiology. 2006;152:2515–28.
- Colonna B, Hofnung M. Rho mutations restore lamB expression in E. coli K12 strains with an inactive malB region. Mol Gen Genet. 1981;184:479–83.
- Chalissery J, Banerjee S, Bandey I, Sen R. Transcription termination defective mutants of Rho: role of different functions of Rho in releasing RNA from the elongation complex. J Mol Biol. 2007;371:855–72.
- Wei RR, Richardson JP. Mutational changes of conserved residues in the Qloop region of transcription factor Rho greatly reduce secondary site RNAbinding. J Mol Biol. 2001;314:1007–15.
- Hinde P, Deighan P, Dorman CJ. Characterization of the detachable Rhodependent transcription terminator of the fimE gene in Escherichia coli K-12. J Bacteriol. 2005;187:8256–66.
- Nystrom T. Growth versus maintenance: a trade-off dictated by RNA polymerase availability and sigma factor competition? Mol Microbiol. 2004; 54:855–62.
- Farewell A, Kvint K, Nystrom T. Negative regulation by RpoS: a case of sigma factor competition. Mol Microbiol. 1998;29:1039–51.
- 100. Ferenci T. What is driving the acquisition of mutS and rpoS polymorphisms in Escherichia coli? Trends Microbiol. 2003;11:457–61.
- King T, Ishihama A, Kori A, Ferenci T. A regulatory trade-off as a source of strain variation in the species Escherichia coli. J Bacteriol. 2004;186:5614–20.
- Notley-McRobb L, King T, Ferenci T. rpoS mutations and loss of general stress resistance in Escherichia coli populations as a consequence of conflict between competing stress responses. J Bacteriol. 2002;184:806–11.
- Versalovic J, Koeuth T, Britton R, Geszvain K, Lupski JR. Conservation and evolution of the rpsU-dnaG-rpoD macromolecular synthesis operon in bacteria. Mol Microbiol. 1993;8:343–55.
- 104. McCloskey D, Xu S, Sandberg TE, Brunk E, Hefner Y, Szubin R, Feist AM, Palsson BO. Multiple optimal phenotypes overcome redox and glycolytic intermediate metabolite imbalances in Escherichia coli pgi knockout evolutions. Appl Environ Microbiol. 2018;84(19):e00823-18.
- Fong SS, Nanchen A, Palsson BO, Sauer U. Latent pathway activation and increased pathway capacity enable Escherichia coli adaptation to loss of key metabolic enzymes. J Biol Chem. 2006;281:8024–33.
- 106. Sauer U. High-throughput phenomics: experimental methods for mapping fluxomes. Curr Opin Biotechnol. 2004;15:58–63.
- Hua Q, Yang C, Baba T, Mori H, Shimizu K. Responses of the central metabolism in Escherichia coli to phosphoglucose isomerase and glucose-6-phosphate dehydrogenase knockouts. J Bacteriol. 2003;185:7053–67.
- Fischer E, Sauer U. Metabolic flux profiling of Escherichia coli mutants in central carbon metabolism using GC-MS. Eur J Biochem. 2003;270:880–91.
- 109. Kimata K, Tanaka Y, Inada T, Aiba H. Expression of the glucose transporter gene, ptsG, is regulated at the mRNA degradation step in response to glycolytic flux in Escherichia coli. EMBO J. 2001;20:3587–95.
- 110. Li Y, Andersson S. The 3-D structural basis for the Pgi genotypic differences in the performance of the butterfly Melitaea cinxia at different temperatures. PLoS One. 2016;11:e0160191.
- Wheat CW, Watt WB, Pollock DD, Schulte PM. From DNA to fitness differences: sequences and structures of adaptive variants of Colias phosphoglucose isomerase (PGI). Mol Biol Evol. 2006;23:499–512.
- 112. Hill NS, Buske PJ, Shi Y, Levin PA. A moonlighting enzyme links Escherichia coli cell size with central metabolism. PLoS Genet. 2013;9:e1003663.
- Schembri MA, Klemm P. Biofilm formation in a hydrodynamic environment by novel fimh variants and ramifications for virulence. Infect Immun. 2001;69:1322–8.
- Hutchins PR, Miller SR. Genomics of variation in nitrogen fixation activity in a population of the thermophilic cyanobacterium Mastigocladus laminosus. ISME J. 2017;11:78–86.
- 115. Hottes AK, Freddolino PL, Khare A, Donnell ZN, Liu JC, Tavazoie S. Bacterial adaptation through loss of function. PLoS Genet. 2013;9:e1003617.
- Venkataram S, Dunn B, Li Y, Agarwala A, Chang J, Ebel ER, Geiler-Samerotte K, Herissant L, Blundell JR, Levy SF, et al. Development of a comprehensive genotype-to-fitness map of adaptation-driving mutations in yeast. Cell. 2016;166:1585–96 e1522.
- 117. Wang L, Spira B, Zhou Z, Feng L, Maharjan RP, Li X, Li F, McKenzie C, Reeves PR, Ferenci T. Divergence involving global regulatory gene mutations in an Escherichia coli population evolving under phosphate limitation. Genome Biol Evol. 2010;2:478–87.
- King T, Seeto S, Ferenci T. Genotype-by-environment interactions influencing the emergence of rpoS mutations in Escherichia coli populations. Genetics. 2006;172:2071–9.

- 119. Phan K, Ferenci T. A design-constraint trade-off underpins the diversity in ecologically important traits in species Escherichia coli. ISME J. 2013;7:2034–43.
- 120. Yang DD, Alexander A, Kinnersley M, Cook E, Caudy A, Rosebrock A, Rosenzweig F. Fitness and productivity increase with ecotypic diversity among Escherichia coli strains that coevolved in a simple, constant environment. Appl Environ Microbiol. 2020;86(8):e00051-20.
- Kumari S, Beatty CM, Browning DF, Busby SJ, Simel EJ, Hovel-Miner G, Wolfe AJ. Regulation of acetyl coenzyme A synthetase in Escherichia coli. J Bacteriol. 2000;182:4173–9.
- 122. Valgepea K, Adamberg K, Nahku R, Lahtvee PJ, Arike L, Vilu R. Systems biology approach reveals that overflow metabolism of acetate in Escherichia coli is triggered by carbon catabolite repression of acetyl-CoA synthetase. BMC Syst Biol. 2010;4:166.
- 123. Tilman D. Resource competition and community structure. Princeton: Princeton University Press; 1982.
- 124. Dykhuizen DE, Dean AM. Predicted fitness changes along an environmental gradient. Evol Ecol. 1994;8:524–41.
- 125. Yao R, Hirose Y, Sarkar D, Nakahigashi K, Ye Q, Shimizu K. Catabolic regulation analysis of Escherichia coli and its crp, mlc, mgsA, pgi and ptsG mutants. Microb Cell Factories. 2011;10:67.
- Knoppel A, Knopp M, Albrecht LM, Lundin E, Lustig U, Nasvall J, Andersson DI. Genetic adaptation to growth under laboratory conditions in Escherichia coli and Salmonella enterica. Front Microbiol. 2018;9:756.
- 127. Lamrabet O, Plumbridge J, Martin M, Lenski RE, Schneider D, Hindre T. Plasticity of promoter-core sequences allows bacteria to compensate for the loss of a key global regulatory gene. Mol Biol Evol. 2019;36:1121–33.
- Blount ZD, Maddamsetti R, Grant NA, Ahmed ST, Jagdish T, Baxter JA, Sommerfeld BA, Tillman A, Moore J, Slonczewski JL, et al. Genomic and phenotypic evolution of Escherichia coli in a novel citrate-only resource environment. Elife. 2020;9:e55414.
- Wannier TM, Kunjapur AM, Rice DP, McDonald MJ, Desai MM, Church GM. Adaptive evolution of genomically recoded Escherichia coli. Proc Natl Acad Sci U S A. 2018;115:3090–5.
- Fogle CA, Nagle JL, Desai MM. Clonal interference, multiple mutations and adaptation in large asexual populations. Genetics. 2008;180:2163–73.
- 131. Buskirk SW, Peace RE, Lang GI: Hitchhiking and epistasis give rise to cohort dynamics in adapting populations. PNAS (USA) 2017;114(31):8330-5.
- 132. Diaz Caballero J, Clark ST, Coburn B, Zhang Y, Wang PW, Donaldson SL, Tullis DE, Yau YC, Waters VJ, Hwang DM, Guttman DS. Selective sweeps and parallel pathoadaptation drive Pseudomonas aeruginosa evolution in the cystic fibrosis lung. MBio. 2015;6:e00981–15.
- Vecerek B, Rajkowitsch L, Sonnleitner E, Schroeder R, Blasi U. The C-terminal domain of Escherichia coli Hfq is required for regulation. Nucleic Acids Res. 2008;36:133–43.
- Takada A, Wachi M, Nagai K. Negative regulatory role of the Escherichia coli hfq gene in cell division. Biochem Biophys Res Commun. 1999;266:579–83.
- Zambrano N, Guichard PP, Bi Y, Cayrol B, Marco S, Arluison V. Involvement of HFq protein in the post-transcriptional regulation of E. coli bacterial cytoskeleton and cell division proteins. Cell Cycle. 2009;8:2470–2.
- 136. Schwartz K, Wenger JW, Dunn B, Sherlock G. APJ1 and GRE3 homologs work in concert to allow growth in xylose in a natural Saccharomyces sensu stricto hybrid yeast. Genetics. 2012;191:621–32.
- 137. Kryazhimskiy S, Rice DP, Jerison ER, Desai MM. Microbial evolution. Global epistasis makes adaptation predictable despite sequence-level stochasticity. Science. 2014;344:1519–22.
- 138. Blount ZD, Barrick JE, Davidson CJ, Lenski RE. Genomic analysis of a key innovation in an experimental Escherichia coli population. Nature. 2012;489:513–8.
- 139. Swofford DL. PAUP*. Phylogenetic analysis using parsimony (*and other methods). 4th ed. Sunderland: Sinauer Associates; 2002.
- 140. Letunic I, Bork P. Interactive Tree Of Life (iTOL): an online tool for phylogenetic tree display and annotation. Bioinformatics. 2007;23:127–8.
- 141. Kinnersley, Schwartz, Yang, Sherlock and Rosenzweig: Evolutionary dynamics and structural consequences of de novo beneficial mutations and mutant lineages arising in a constant environment. NCBI BioProjects, accession number PRJMA517527. https://www.ncbi.nlm.nih.gov/bioproject/PRJNA517527. Access date 28-Jan-2019.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.